Author's response to reviews

Title: PRospective Observational POLIsh Study on Post-Stroke Delirium (PROPOLIS): methodology of hospital-based cohort study on delirium prevalence, predictors and diagnostic tools.

Authors:

ELZBIETA KLIMIEC (eklimiec@gmail.com)
TOMASZ DZEDZIC (dziedzic@su.krakow.pl)
KATARZYNA KOWALSKA (katrzyna.olga.kowalska@gekail.com)
ALEKSANDRA SZYPER (ola.szyper@gmail.com)
JOANNA PERA (pera@su.krakow.pl)
PAULINA POTOCZEK (geromed@geromed.pl)
AGNIESZKA SLOWIK (slowik@cm-uj.krakow.pl)
aLEKSANDRA KLIMKOWICZ-MROWIEC (Aleksandra.Klimkowicz@mp.pl)

Version: 2 Date: 31 May 2015

Author's response to reviews:

Dear Editor
Timothy Shipley
Executive Editor
BMC Neurology

Dear Editor,

I enclose a revised version of manuscript entitled “PRospective Observational POLIsh Study on Post-Stroke Delirium (PROPOLIS): methodology of hospital-based cohort study on delirium prevalence, predictors and diagnostic tools” to be consider for publication in the BMC Neurology as a Study protocol. I added the line and page numbers to this version.

This study has been peer reviewed by its funding body.